Author’s response to reviews

Title: Usefulness of Occupation and Industry information in Mortality Data in South Africa from 2006 to 2015

Authors:

Kerry Wilson (KerryW@nioh.ac.za)

Nisha Naicker (NishaN@nioh.ac.za)

Tahira Kootbodien (TahiraK@nioh.ac.za)

Vusi Ntlebi (VusiNT@nioh.ac.za)

Felix Made (FelixM@nioh.ac.za)

Nonhlanhla Tlotleng (NonhlanhlaT@nioh.ac.za)

Version: 2 Date: 10 Jun 2019

Author’s response to reviews:

a reference to Figure 2 was included in the text in the results section line 161.